

DOI: 10.32604/cmc.2024.048893

REVIEW





### AI-Driven Learning Management Systems: Modern Developments, Challenges and Future Trends during the Age of ChatGPT

### Sameer Qazi<sup>1,\*</sup>, Muhammad Bilal Kadri<sup>2</sup>, Muhammad Naveed<sup>1,\*</sup>, Bilal A. Khawaja<sup>3</sup>, Sohaib Zia Khan<sup>4</sup>, Muhammad Mansoor Alam<sup>5,6,7</sup> and Mazliham Mohd Su'ud<sup>6</sup>

<sup>1</sup>Department of Computer Science, Faculty of Engineering, Science and Technology, IQRA University, Karachi, 75500, Pakistan

<sup>2</sup>College of Computer and Information Sciences, Prince Sultan University, Riyadh, 11586, Saudi Arabia

<sup>3</sup>Faculty of Engineering, Department of Electrical Engineering, Islamic University of Madinah, P.O. BOX 170, Madinah, 41411, Saudi Arabia

<sup>4</sup>Faculty of Engineering, Department of Mechanical Engineering, Islamic University of Madinah, P.O. BOX 170, Madinah, 41411, Saudi Arabia

<sup>5</sup>Faculty of Computing, Riphah International University, Islamabad, 46000, Pakistan

<sup>6</sup>Faculty of Computing and Informatics (FCI) Multimedia University, Cyberjaya, 63100, Malaysia

<sup>7</sup>Faculty of Engineering and Information Technology, School of Computer Science, University of Technology Sydney, Sydney, NSW 2007, Australia

\*Corresponding Authors: Sameer Qazi. Email: sameer.hashmat@iqra.edu.pk; Muhammad Naveed. Email: shmnaveed@hotmail.com

Received: 21 December 2023 Accepted: 08 April 2024 Published: 15 August 2024

#### ABSTRACT

COVID-19 pandemic restrictions limited all social activities to curtail the spread of the virus. The foremost and most prime sector among those affected were schools, colleges, and universities. The education system of entire nations had shifted to online education during this time. Many shortcomings of Learning Management Systems (LMSs) were detected to support education in an online mode that spawned the research in Artificial Intelligence (AI) based tools that are being developed by the research community to improve the effectiveness of LMSs. This paper presents a detailed survey of the different enhancements to LMSs, which are led by key advances in the area of AI to enhance the real-time and non-real-time user experience. The AI-based enhancements proposed to the LMSs start from the Application layer and Presentation layer in the form of flipped classroom models for the efficient learning environment and appropriately designed UI/UX for efficient utilization of LMS utilities and resources, including AI-based chatbots. Session layer enhancements are also required, such as AI-based online proctoring and user authentication using Biometrics. These extend to the Transport layer to support real-time and rate adaptive encrypted video transmission for user security/privacy and satisfactory working of AI-algorithms. It also needs the support of the Networking layer for IP-based geolocation features, the Virtual Private Network (VPN) feature, and the support of Software-Defined Networks (SDN) for optimum Quality of Service (QoS). Finally, in addition to these, non-real-time user experience is enhanced by other AI-based enhancements such as Plagiarism detection algorithms and Data Analytics.

#### **KEYWORDS**

Learning management systems; chatbots; ChatGPT; online education; Internet of Things (IoT); artificial intelligence (AI); convolutional neural networks; natural language processing



#### 1 Introduction

During the years 2019–2020, the problem of COVID-19 suddenly engulfed all social activities, particularly education, to contain the pandemic disease [1]. The foremost and most prime sectors among those affected are schools, colleges, and universities. The education system of entire nations had shifted to online education during this time. The Learning Management Systems (LMSs) in universities, which previously functioned in supporting roles only, suddenly found themselves thrust in the forefront [1-3]. For instance, in previous years, LMSs had limited features and functions. They lacked analytical abilities; the User Interface was far from being user-friendly, and they could not host safe exams. This precipitated a race to enhance and improve LMSs in universities to entirely support online education when physical classes were impossible. Consequently, today, LMSs are synonymous with education, offering end-to-end solutions replete with unique features and functionalities. The global LMS market is projected to grow from USD 13.38 billion in 2021 to USD 44.49 billion in 2028 [2]. To date, online exams software that can combat the menace of cheating has only been effectively developed for the Multiple Choice Questions (MCQ) system of exams [4] but not for other varied natures of exams, e.g., subjective, theoretical, numerical-based with varied guidelines such as open/closed book and internet permissions. Due to these reasons, both schools'/colleges'/universities' academics and students'/parents' [5] are dissatisfied with the online classes and online examination systems. This dissatisfaction primarily results from the (i) Lack of proper teacher-student interaction during online classes due to various internet and electricity disruptions; (ii) Ease of cheating in online tests and exams due to students using other electronic gadgets and social media software or sitting in close proximity to each other outside of camera detection range; (iii) Privacy issues of female teachers and students in turning on the camera for classes and exams; (iv) For non-MCQ based subjective exams, time management issues by students, especially in the conversion of written scripts to a form suitable for electronic submission, e.g., scanning of the document requires significant time to be utilized from within exam duration and online submission also becomes tricky due to load on university LMS as the traffic spikes suddenly; and (v) Internet disruptions.

Similarly, during the pandemic outbreak [1], for online class conduction, educational institutes have been favoring either licensed Video Conferencing Systems (VCSs), e.g., Microsoft Teams [6], or freely available VCSs, e.g., Zoom [7]. These VCSs were developed with the prime focus of online meetings and never with the intention of conducting educational classes and exams. Although some specialized online examination modules are developed by Pearson VUE [8], Autoproctor [9], ProctorU [10], etc., there is still a need for the unification of various tools, which have been developed along with recent research studies, to create an online class and exam conducting system that can be seamlessly integrated with the university LMS.

Future LMS will be equipped with AI-based tools for the adaptive learning experience, online proctoring, advanced data analytics and intelligent user session management. These will in turn fuel research into several cross-disciplinary research such as faster hardware technologies to support advanced image and video processing and analytics, user biometrics-based identification/authentication. In addition, they will also require faster, reliable, secure data delivery over the Internet with the help of Software Defined Networks (SDN), advanced data encryption and elastic data applications to support real time, computationally intensive AI-algorithms running at back end.

In addition to these technical challenges, LMS will also be challenged in future with other ethical challenges due to AI based content generation tools that the student will consult for completing assignments. With the recent breakthroughs in Generative-AI technologies like Open AIs ChatGPT and Google Bard, there has been an increasing apprehension by academics that chatbots interfaces provided these technologies utilizing smart AI-algorithms can facilitate a student in plagiarizing

assignments that cannot be detected by conventional plagiarism detection tools like TurnItIn [11], IEEE CrossCheck [12], and iThenticate [13]. McIntire et al. [14] used a technique to study the statistical patterns of word frequency in messages and inter-message intervals to detect if student response was indeed generated by a chatbot or not. More recently, several researchers have developed online tools to detect whether responses submitted by students are original work or generated by chatbots such as ChatGPT [15–19]. Thus, one requirement of advanced plagiarism detection software will be to counter AI-algorithms to detect the involvement of a chatbot.

Thus summarizing, LMS development for next generation education delivery will not be limited to just one particular area of research rather it will affect several layers of the OSI protocol stack for networked LMS applications. The main contribution of our work will be to present a comprehensive survey of Artificial Intelligence (AI) based novel techniques as proposed in the recent research literature, e.g., [9–11] in the development of LMSs. These techniques work in the Application, Presentation, Session, Transport, and Network layer in the function of LMSs. This comprehensive survey covers (i) AI-enabled adaptive learning techniques to tailor the course according to individual student needs, (ii) Techniques used for extracting student bioinformatics (fingerprints, face detection, etc.) to authenticate using AI support, (iii) Making the online exams resilient to internet outages, (iv) Automated video surveillance of students attempting exams (eye gaze and face motion tracking) (v) AI-based tools for plagiarism detection (chatbot generated responses or use of prohibited URLs) and (vi) the overall system being more student-friendly and privacy-preserving especially for female students and teachers.

The roadmap of the paper (Fig. 1) will be as follows: Section 2 will describe related work; Section 3 covers the recent developmental trends and research for AI-backed LMSs; Section 4 will present existing research challenges. Section 5 discusses the future trends on the development of LMSs in the presence of generative-AI based chatbots like ChatGPT. Then, Section 6 provides a discussion on the different aspects discussed in the previous sections of this paper. Finally, Section 7 concludes this paper.



Figure 1: Roadmap of the paper

#### 2 Related Work

With the outbreak of the COVID-19 pandemic, the entire world switched from physical education to an online mode of education. This precipitated the change in the development of effective online educational tools for use in education institutes. Several researchers have investigated the shift in this paradigm [12]. Murad et al. [20] conducted a survey for teacher readiness for the online education system in Indonesia. Mondego et al. [21] conducted a similar study but focused primarily on online teaching techniques that could be used by academicians at higher education institutes. Suprivation et al. [22] stressed the need for effective teacher training for online educational tools and the institutes to constantly monitor student progress and satisfaction levels. Similarly, researchers have conducted an exhaustive literature review of different strategies and practices adopted by different educational institutes and challenges faced by academics and students in the new online education delivery method. Other researchers have used Twitter data to garner popular public opinion and feedback related to online education during the COVID-19 pandemic. Again, different regulatory processes and effective communication tools were found to be major impediments to ensuring the quality of online education delivery methods. Edy et al. [23] discussed the impact of online education through video conferencing tools such as Zoom, WebEx, Meet, etc., on vocational education, which requires hands-on training by the students. The study found that there were remarkable shortcomings in students' learning outcomes. Salas-Pilco conducted a similar study for STEM education in Latin America [24]. Wang et al. [25] conducted a SWOT analysis for college-level online education.

The use of AI in every aspect of our lives has exponentially increased during the past few years [26,27]. AI systems, as the name suggests, are the systems that are trained using large amounts of data to mimic human brain-like thinking processes and perform routine tasks [28,29]. AI has revolutionized the way we used to perform our day-to-day tasks and has found many applications in education, healthcare, banking, transportation, entertainment, and smart-systems based on Internet of Things (IoT) technology, etc. [30,31]. Al-Chalabi et al. [32] used adaptive learning based on a decision tree algorithm to train students using e-learning tools.

Nenkov et al. [33] developed an AI-based personalized learning system. This system uses agents in the form of AI-powered chatbots to automate the interaction between the student and the teacher using an LMS based on the Moodle platform. Skrebeca et al. [34] considered a very novel approach to automating student-teacher interaction by using lecture delivery methods such as chat messages. Albayrak et al. [35] considered the various applications of AI chatbots in various sectors, such as education, telecommunication, banking, health, customer call centers, and e-commerce. Deveci Topal et al. [36] conducted an experimental online course delivered via a chatbot application for 5th grade science course; although results about their performance enhancement in the course remained inconclusive, the students rated the learning experience via chatbot application as superior. However, Yang et al. [37] reported significant improvement in medical students' suturing and ligature training through the use of AI-assisted tutoring in conjunction with clinical studies. Krishnan [38] and Schwendicke et al. [39] have advocated the use of AI-based modern learning techniques in the field of Oral and Maxillofacial surgery and dentistry.

Rahman et al. [40] studied the technical programming challenges for several cloud-based chatbots such as IBM Watson, Microsoft bot, AWS Lambda, Heroku, and many others. OpenAI has quite recently introduced a conversational chatbot popularly called ChatGPT [41]. This chatbot is developed on the Generative Pre-trained Transformer (GPT) language model, which is precisely trained to produce human-like output in response to user inputs in natural language [42]. Deep learning and natural language processing techniques are used to train ChatGPT on massive amounts of text data

to understand human language [43]. ChatGPT is the latest state-of-the-art chatbot from OpenAI, which has shown that its system can generate passing grade-level responses to several graduate-level bar and medical examinations [44]. King et al. [45] included ChatGPT as a co-author in their journal article, while ironically, other research publications disapproved of using chatbots as a co-author.

Ranoliya et al. [46] considered the problem of AI-based chatbots for answering student FAQs about a university. To address this problem, in this paper, the authors propose the design of a chatbot using AI-Markup Language (AIML) and Latent Semantic Analysis (LSA). Ugurlu et al. [47] tackled the same problem by designing an AI-based virtual assistant through a chatbot for a Turkish University. Wang [48] discussed the fusion of AI and virtual reality technology for university-level physical education.

Chiang et al. [49] used deep learning algorithms for the prediction of student performance in introductory computer programming courses using Moodle-based LMS log data. Aljaloud et al. [50] conducted a similar study by employing a combination of convolutional neural networks (CNNs) and long short-term memory (CNN-LSTM) to predict student information using LMS data. Another study carried out by Akmese et al. [51], in which the author found that student academic performance prediction based on their social and demographic situation could be effectively predicted through the Random Forest Machine Learning algorithm.

Slusky [52] presented a comprehensive literature survey on several commercially available online examination systems developed by PearsonVUE [8] ProctorU [10], Eklavya [53], Respondus [54], Talview [55], etc., and used an online proctoring system to prevent cheating. Requirements and capabilities of various commercially available online exam proctoring systems have been critically analyzed by Nigam et al. [56].

Chiu et al. [57] considered the application of AI to the field of education in four key educational domains, i.e., learning, teaching, assessment, and administration, and presented a thorough bibliographical survey. Salas-Pilco [58] considered the effects brought about by AI and robotics on the intellectual learning outcomes of students in higher education.

Vahabzadeh et al. [59] developed AI-based smart glasses for the learning experience of autistic students as a tool for their social and emotional behavioral learning aid. Another study carried out by Standen et al. [60] discusses the use of AI-algorithms for adaptive learning approaches for individuals with learning disabilities. McCarthy et al. [61] developed an AI-based tutor for teaching braille language to blind individuals.

Sun et al. [62] and Chen et al. [63] discussed the applications of blockchain technology in the field of online education for secure storage of learning records, course credibility, credit and certificate certification, student privacy, and course sharing. Blockchain technology [62,63] can store such data in a trusted, distributed manner, provide credible digital certificates, realize learning resource sharing with smart contracts, and protect intellectual property through data encryption.

Villegas-Ch. et al. [64] utilized an AI-model and implemented it through a Hadoop framework for efficient prediction of learning activities tailored to the needs of all students. The AI-system processes the student's past performance information and recommends activities that focus on each student's abilities and needs. The practical integration of these systems into universities' LMSs can create an adaptive educational model that responds to the new challenges of society. Sharma et al. [65] also applied a machine learning approach to provide students with timely, actionable feedback to improve their academic performance in a futuristic adaptive learning environment. Tsai et al. [66] provided a similar solution using a Deep Learning model.

#### 3 Recent Developments Towards Making Smart LMSs for Online Education Delivery

The requirements of an effective online LMS are multifaceted. It should incorporate an AIenabled adaptive learning approach, accurate student authentication, and online proctoring algorithms, amongst other features. Currently, the universities start with a video conference application program (either licensed or free) and tailor their exams to work around it. It is necessary to make a secure, online class conduction and exam system at the forefront of the design, which will have both hardware and software components to authenticate the exam candidate and prevent cheating or other unethical means. The proposed architecture is shown in Fig. 2.



Figure 2: Overall semantics of an AI-integrated LMS architecture

### 3.1 Improved Graphical User Interfaces–(Application|Presentation Layer Enhancement)

Modern Learning Management System uses intuitive graphical user interfaces so students can easily interact with it. The user interface includes icons, buttons, menus, search bars and also provision for online tools through hyperlinked third-party software tools. This greatly speeds up the learning interface for students and faculty alike. Through improved user interfaces, learning management system (LMS) provide easy-to-use tools for the administration, documentation, tracking, reporting, automation, and delivery of educational courses, training programs, materials or learning and development programs.

#### 3.2 Development of an Examination System Integrated with LMS–(Session Layer Enhancement)

Currently, the online examination systems supported by open-source LMS of universities can only support a variety of online exam types, e.g., Subjective, Objective, and MCQs. However, the prevention of unfair means in attempting the online exam can be partially guaranteed for MCQ-based exams through measures such as time limits on a question and scrambling of questions and answer choices

for candidates, and that too is only applicable when students attempt the exam in specially controlled conditions, e.g., a dedicated exam center only. In other types of non-MCQ-based exams, i.e., numerical-based, subjective, or theoretical, limited techniques have been developed for Massive Open Online Courses (MOOCs), so it is easier for the candidates to use unfair means in online exams on this pattern.

Under the new Outcome Based Education (OBE) systems [1,3], it is mandatory to judge the candidate's performance on non-MCQ-based exams to measure their course learning outcomes (CLOs) and program learning outcomes (PLOs) [3]. To alleviate this problem of unavailability of videobased online proctoring in LMSs based on the open-source, popular Moodle architecture, usually 3rd party VCSs are used for online proctoring, while the exam paper and completed exam may still continue to be shared through the official LMS channel. As a consequence, (1) The student has to open two software simultaneously to be able to attempt the online exam; (2) As these 3rd party software are not integrated with the LMS, this effectively creates a hurdle in automated solutions for proctoring, a human still has to manually watch the video feeds of all students. Hence, many automated tasks like face recognition and handwriting match, which could easily be performed if these systems had access to LMS data, are not possible.

Many recently developed 3rd party online examination tutoring software are providing additional support for seamless integration with current learning management platforms of universities like Moodle [67], Canvas [68], and Blackboard [69] through open-source customizable plugins. These include ProctorFree [70], HonorLock [71], and Mercer-Mettl [72].

### 3.3 AI-Surveillance Algorithms during Conduction of Regular Classes and Online Exams–(Session Layer Enhancement)

One of the significant issues identified by university academics is that while taking virtual online classes through any video conferencing systems, due to the absence of face-to-face contact, it is often difficult to track which student is present and actively participating in the lecture or merely logged into the VCS. Videos are often switched off due to privacy concerns, especially for female students or when the internet bandwidth is low. A more dire situation may be another proxy student attending the lecture and attempting quizzes/exams on behalf of an enrolled student.

Similarly, during the conduct of the exams, it is difficult to detect any means of unfair cheating by the students. In fact, the use of unfair means is common, especially for subjective exams. Tarigan et al. [73] conducted an exhaustive literature review for different types of academic misconduct by the students attempting online exams. Rabiha et al. [74], in their literature review, drew the conclusion that the most popular online cheating detection techniques used in popular literature were facial motion and head pose, followed by eye movement or gaze detection. Finally, in the third place were the facial movements and gaze detection for multiple targets, as depicted in Fig. 3.

Automated AI-algorithms will be able to establish the identity of the candidate through advanced Face Recognition algorithms. Similarly, Advanced Handwriting checks on answer scripts submitted by students, which is not possible to be done manually by teachers, given a large number of students. This may be managed by implementing CNN algorithms to implement AI-based surveillance of students attending lectures, submitting assignments throughout semesters, and online examination invigilation.

The CNN algorithm architecture is shown in Fig. 4. In this architecture, during the normal conduction of semester classes, student pictures, and their handwriting samples may initially be pulled from the LMS database as a first instance of a labeled dataset sample. More training data of student pictures and handwriting samples can be generated during the normal progress of the semester to allow the CNN to train up to an acceptable level before its implementation can be allowed with low false

positives and false negatives. Initially, to validate this training dataset, occasionally, the CNN will take manual help from the professors to limit any errors in the training dataset. In a standard 50-min lecture duration, a considerable amount of labeled datasets for visual student recognition can be generated.



Figure 3: Common features of online proctoring solutions





Similarly, ample amounts of handwriting samples can be obtained from student assignments/quizzes being submitted through university LMS systems. In advanced stages, such student

data can also be carried forward to subsequent semesters. The above architecture can determine with high accuracy if a student's identity can be established reliably from the match of this video/image data and handwriting samples. An anomaly will be appropriately flagged for manual investigation by the college administration. Also, valuable statistics can be gathered about the learning experience of students through online education [75] and an evaluation system [76].

Li et al. [77] developed a multi-index cheating detection technique using an LSTM neural network. Similarly, Samir et al. [78] and Essahraui et al. [79] developed a cheating detection system for onsite as well as online examinations by introducing novel computer vision techniques based on CNN-Facial-Landmarks and LSTM models for the classification, which incorporate the detection of students' head postures and hand movements during the conduction of an online exam. The algorithm can detect cheating for both single-user and multi-user tracking scenarios.

Yulita et al. [80] developed an efficient classifier based on the combination of AdaBoost and SVM for cheating detection using datasets of human activity detection. They found that the proposed approach was more efficient in the detection of cheating compared to other machine-learning techniques. Li et al.[81] developed an online proctoring application by using two cameras working in tandem for 3D gaze tracking. They developed novel gaze motion formulas and also trained a custom-built object detection classifier to detect cheating targets in exams.

Dilini et al. [82] used a One-Class Support Vector Machine (OCSVM) technique for student gazebased cheating detection. Gupta et al. [83] have developed a smart Bluetooth-based camera application for online exam cheating detection even if the student is taking the exam at his home using the Deep Learning approach. Rudian et al. [84] indicated security vulnerabilities for digital content generation in online lecture delivery and exam creation using H5P, where often the answer to a question can be easily obtained using the source code itself used to develop the exam. They suggested useful measures that can be used by the instructors in setting the question papers in these situations.

Kasliwal [85] used an open-source packet-capturing application, KISMET, for packet sniffing at the student's computer to see if the student used any of the prohibited URLs during the exam. Bawarith et al. [86] used student biometric authentication via fingerprints, eye tracking, and motion detection techniques via embedded hardware to determine any suspicious activity by the student attempting the online exam.

Atoum et al. [87] developed a multimedia analytics system that continuously tracks six features for the duration of the online exam; gaze detection, text detection, voice detection, phone detection, active window detection, and user verification.

There are a few AI-based software that also help bridge the gap between online lectures and online examinations. One such software is Examus, which monitors students' behavioral characteristics during online lectures and provides them with proctoring services for better monitoring in online exams [52]. Table 1 provides a summary of research in this domain utilizing AI-based smart algorithms.

Research work or commercially available proprietary software	Technique for prevention of academic misconduct in online examination	Artificial intelligence algorithm
Talview [55]	Synchronous and asynchronous use of recorded audio-video streaming and screen snapshots capturing user's computer activities	Proprietary video tracking software
Yulita et al. [80]	Machine learning classification algorithm (on datasets) for human activity detection	Adaboost + SVM
Li et al. [81]	Two camera applications for 3D gaze tracking and cheating target detection	Custom-built gaze motion formulas and cheating target detection based on custom-built object detection classifier
Dilini et al. [82]	Eye gaze detection	One class SVM (OCSVM)
Gupta et al. [83]	Smart Bluetooth-based camera application	Deep learning technique
Samir et al. [78] and Essahraui et al. [79]	Head posture and hand movements	CNN-facial-landmark, media pipe models. And LSTM classification model for cheating case prediction
Atoum et al. [87]	Voice detection, user verification, text detection, gaze estimation	Multimedia analytics
Rudian et al. [84]	Security vulnerabilities of software used in the development of online exams such as H5P	NA
Kumar et al. [88]	Handwritten character recognition for writer identification	RNN and LSTM, mixture density networks, reinforcement learning, generative adversarial networks
KreB et al. [89]	Real-time handwritten character recognition with a in pen interface	Deep neural networks (DNN) employing connectionist temporal classification (CTC) algorithm and language models to the decoding algorithm

**Table 1:** Summary of research works in the domain of artificial intelligence for academic misconduct prevention in online exams

(Continued)

Table 1 (continued)				
Research work or commercially available proprietary software	Technique for prevention of academic misconduct in online examination	Artificial intelligence algorithm		
Rajesh et al. [90]	Handwritten character recognition for conversion to text suitable for plagiarism detection	Convolutional neural network (CNN) and Bi-Directional long short-term memory (BiLSTM) based recurrent neural network (RNN)		
Aljaloud et al. [50]/ Akmese et al. [51]	Student performance prediction in online courses and exams	CNN + LSTM applied on blackboard/Random forest algorithm		
Pareira et al. [91]	AI-based recommender system for POJ (Programming online judges) to select questions for assignments to avoid plagiarism and detect student efforts in completing the assignment	Custom built AI-based recommender system		
McIntire et al. [14]	Chatbot-generated response detection	Words per message, inter message delays		
GPT zero [15] Open AI GPT2 output detector [16]	Chatbot-generated response detection Chatbot-generated response detection	Not mentioned Transformer model		
ContentScale AI content detector [18], writers AI content detection [19]	Chatbot-generated response detection	Not mentioned		
Kasliwal [85]	Packet capturing and detection of prohibited URLs	KISMET open-source packet		
Bawarith et al. [86]	Authentication using student fingerprints and continuous eye tracking and motion tracking	Embedded hardware digital persona U. Are.U 4500 HD USB fingerprint reader, and eve tribe tracker		
Respondus [54]	Respondus LockDown browser keeps open only one browser and disables access to the rest of the endpoint (user computer) environment. Respondus monitor checks students' behavior by using a webcam and performs video analytics to detect behavioral events that can signal cheating during online exams	Respondus LockDown browser and respondus monitor proprietary software		

(Continued)

Fable 1 (continued)				
Research work or commercially available proprietary software	Technique for prevention of academic misconduct in online examination	Artificial intelligence algorithm		
BeyondTrust privilege management [92]	Secure administrative privileges and access management for the examinee's computer	Proprietary software		

#### 3.4 Digital-Stationery Connected by IoT Concept-(Network Layer Enhancement)

Another challenging issue faced by teachers is that students may submit work completed by another student, passing it off as their own. Sometimes, images of handwritten work are not legible. In this solution, students will attempt the exam using Digital Writing pads, which will make the job of handwriting recognition simpler and faster. Also, such digital stationery will credibly establish the real-time geographic location of the exam candidate throughout the duration of the exam, even in the absence of video surveillance. Liwicki et al. [93] made the case for handwriting detection through pen-enabled displays/tablets for instant handwriting recognition.

AI-algorithms can then be used for real-time detection of handwriting. Similarly, Kumar et al. [88] made a case for using an AI-algorithm for handwritten character recognition for writer identification using Recurrent Neural Networks (RNN) and LSTM networks. KreB et al. [89] proposed a Deep Neural Network (DNN) based technique in which sensor data is first stored in a compressed format in the electronic pen through quantization of weights of the DNN layers. Then, a Connectionist Temporal Classification (CTC) algorithm is used to minimize the link bandwidth. This allows transmitting only a small matrix from the pen to the mobile device on which the decoding is performed. Additionally, to further improve the accuracy of handwriting recognition, the authors added a Language Model (LM) to the decoding algorithm executed on the mobile device. This technique distributes the load on the hardware while making edge computing possible for handwriting recognition.

#### 3.5 Improved Security Features-(Transport and Network Layer Enhancements)

Respondus [54], a commercially available software that uses Respondus LockDown Browser, keeps open only one browser and disables access to the rest of the endpoint (user computer) environment. Respondus can monitor students' behavior by using a webcam and perform video analytics to detect behavioral events that can signal cheating during online exams. Similarly, BeyondTrust Privilege Management [92], another commercially available software, is an endpoint security software that combines privilege management, application control, and sandboxing. It helps prevent unnecessary granting of admin rights to users and compromising security by enabling users to log on to the desired applications using their standard user accounts. This is done by limiting traffic to only applications using specified TCP port.

Another but limited solution for securing access control is a virtual private network (VPN). Some already available applications like Sonicwall [94] enable restricted and secure VPN access and firewall solutions for higher education examination purposes.

#### **4** Current Challenges

#### 4.1 AI-Based Automated Proctoring: Reliable? (Session Layer Enhancement)

There is an increasing concern about the shift of control from academic institutions to commercial providers, as investigated by Selwyn et al. [95]. The AI-algorithms can frequently make mistakes or out-of-context decisions as they are only as good as they are 'trained.' Henry et al. [96] brought to light a case of a student who was disqualified from an online exam by an online proctoring algorithm, only to be reinstated swiftly when she lodged a formal appeal against the decision, with the professor apologizing for the mistake made by the software. There is a dire need to establish the amount of autonomy academic institutions are willing to give to automated algorithms for all academic misconduct during online examinations while exonerating their own faculty members completely if any student is unjustly marginalized. Also, there is a need for special grievance policies should a student decide to contest a decision made by an automated AI-algorithm.

#### 4.2 Development of AI-Based Computer Adaptive Algorithms to Cope with InternetlPower Outages-(Application/Session Layer Enhancement)

Computer adaptive testing was established three decades ago by international testing services like ETS for GRE exams. The questions appear from different difficulty level pools and try to zero in on the true academic potential of the student. Newer initiatives [97] follow a two-pronged approach for harnessing AI in effective learning of students using both adaptive and adaptable learning. Another issue that can be solved using this approach is that conventional LMS systems based on Moodle architecture only run the exam timer on the main LMS server. Sometimes, internet disruption during time-limited exams proves disastrous for candidate's performance in the exam when they miss valuable examination time in the event of client-side computer problems such as internet or power outages. It is necessary to have a seamless examination system in which the adapted examination system can run locally on the client side and an encrypted version of the past time-stamped student responses be saved on the local computer with a local timer operating on the local (client side) computer with candidate's browser and audio/video logs which could be submitted as soon as Internet service is restored.

#### 4.3 Improved Privacy Features Using AI–(Application Layer Enhancement)

Female teachers and students are reluctant to turn on their videos during the conduction of online classes and exams due to privacy reasons that any videos recorded in the class or during exam invigilation could be used by cybercriminals. Due to this, many female teachers opt for an asynchronous mode of online classes, or the conduction of live classes with the video turned off. This creates learning issues for many students. Similarly, female students are reluctant to have their video recorded while conducting online exams. Thus, privacy issues are at the forefront for both female teachers and students, as highlighted in recent research [98]. Student has frequently expressed concerns about online proctoring systems recording private and sensitive information from their computers [99,100]. Online education systems should safeguard these concerns so that female teachers and students are able to share their videos with their privacy concerns addressed using encrypted video streams [101–103] and advanced AI-based algorithms to still be able to carry out their face detection accurately even with the presence of hijab/niqab [104–106]. There is also an increasing concern about ethics when handling students' and teachers' private information; this has been discussed in detail by Henry et al. [96].

#### 4.4 Improved Resilience to Heterogeneous Internet Speeds-(Transport Layer Enhancement)

The other important issue is that the students connecting to such online video conferencing systems connect through different networks with different connection speeds and bandwidths. Turning on high-resolution videos for multicasting puts needless strain on the network, and often, universities advise teachers and students to keep their videos turned off. This again exacerbates the issue of students using unfair means during regular classes and exams. Top automated proctoring solutions commercially require high bandwidth internet connections for satisfactory performance. Pearson VUE online proctoring software requires at least a 3 Mbps connection [107]. Elastic applications with rate adaptation according to network conditions have been developed using efficient transport layer algorithms, as shown in Fig. 5.



Figure 5: Rate-adaptive data streaming

Notable works in rate-adaptive video streaming applications include Dynamic Adaptive Streaming over HTTP (DASH) [108], and HTTP Live Streaming protocol (HLS) [109], which adapt the bitrate based on network conditions. DASH divides videos into segments and applies H.264 [102] or HEVC advanced video coding [110] to encode them at varying quality levels (bit-rates). These encoded segments are then stored on the server. The client selects the most suitable segment to transfer the video based on the current network capacity estimation. Another viable option for real-time video streaming is by using Software Defined Networks (SDN). Service providers can interact with apps and devices through a programmable interface provided through SDN. This gives consumers the ability to negotiate bitrate streams for real-time services and make quick requests for resource allocation for various services. The SDN design offers a method for separating network control and routing duties, leading to programmable and abstracted control that is unrelated to network services and applications. SDN makes it feasible to design flexible topologies and new protocol extensions for networks. Only a few emulators, including Mininet [111] and OpenVswitch [112] are readily available right now.

Like rate-adaptive data transmission schemes to salvage real-time data transmission in the face of internet outages, legacy data compression may also be used, albeit for non-real-time or near real-time applications. KreB et al. [89] proposed a handwriting recognition technique using an electronic pen in a deep neural network (DNN) technique in which sensor data is stored in a compressed format. This allows transmitting only a small matrix from the pen to the mobile device on which the decoding is performed. Similarly, Rajesh et al. [90], in their work, HWRCNet, developed a data-compression-based handwritten characters recognition technique combining CNN and Bi-Directional LSTM (BiLSTM) based RNN.

#### 4.5 AI-Algorithm Implementation at Edge–(Session Layer Enhancement)

It is also important to implement AI-algorithms at the edge through the IoT concept [113], which will make the system secure [114], and scalable and reduce the burden on a centralized AI inference

engine for real-time cheating detection. Online proctoring is heavily dependent on real-time computer vision algorithms. These advanced computer vision algorithms run on the server-end. In the event of session disconnection due to a real or doctored internet/power outage, the computer vision processing will cease giving an opportunity to students for academic misconduct. If these algorithms are running at the edge, i.e., client-end, then automated proctoring can continue unabated and also notify the central server if any anomaly is detected at any point in time.

### 4.6 Use of AI for Adaptive Learning through Flipped Classroom Model Implementation in LMS–(Application|Presentation Layer Enhancement)

Varnavsky [115] investigated the efficiency of Flipped Classroom technology using chatbot-based student interaction. Pareira et al. [91] devised an intelligent system where an AI-based recommender system proposes tasks for student assignments in computer programming courses to avoid plagiarism and truly reflect students' efforts by monitoring the IDE used to develop the assignment. Initiatives like [97] follow a two-pronged approach to harnessing AI in the effective learning of students. AI-based adaptable and adaptive learning is possible through both a mobile and web interface. A learner can learn what he is capable of learning (adaptive) and in the manner he wishes (adaptable learning). The system, through deep data analytics, determines the contents of the course and the manner in which it should be presented to enhance the learning of the student as per previous attempts in completing assignments and tests. An intuitive UIUX design is also a must to support enhanced user experience.

#### 5 Future Trends on the Development of Learning Management Systems

## 5.1 Permitted Use of Generative-AI as an Acceptable Learning Augmentation Tool–(Application Layer Enhancement)

Future education delivery including that using LMS, will be highly affected by generative-AI based content generation and knowledge discovery tools. The popularly available ChatGPT AIchatbot to everyone including students is trained on a vast amount of data available from the Internet, and its GPT technology allows it to produce human-like responses to various input prompts. ChatGPT is an advanced language model that utilizes the "transformer architecture" to perform various natural language processing tasks such as language understanding and generation [116]. This unique architecture enables AI-powered chatbots to interpret the relationship between words in a sentence, thereby preserving context and producing responses that are both logical and coherent [117]. In engineering, sciences, and related research disciplines, this leads to technical content generation like experimental protocols, designs, and results [118]. When it comes to research content writing, the ChatGPT-based approach offers a significant advantage in terms of efficiency. Writing papers or a thesis can be a time-consuming task, but with this method, users can save time and effort by receiving instantaneous relevant responses from ChatGPT based on specific input prompts related to their research areas. Technical knowledge and area expertise are crucial for engineering and scientific research paper writing, which can make it quite challenging. However, the use of ChatGPT AI-chatbot can be of great help, as it can generate relevant content based on carefully drafted prompts provided by the researcher, thus saving time and effort in the writing process [119]. Another advantage of utilizing ChatGPT-based AI-chatbot for scientific and engineering research paper writing is the potential to enhance the consistency and precision of the way the papers are written. Publishers like IEEE, Springer, Elsevier, Wiley, etc., who publish technical research papers generally demand a high level of accuracy and consistency from the authors, which can be difficult for human writers (especially those who do not have English as their first language) to maintain consistently. ChatGPT-based techniques can produce a dependable and precise output, decreasing the likelihood of errors and disparities [120].

However, ChatGPT also has some limitations. One of the major drawbacks of the ChatGPT model is its lack of deep understanding and potential to produce erroneous or faulty information. ChatGPT is trained on a vast amount of data (from the internet) that could contain inaccuracies or biases, resulting in the model generating misleading information. Furthermore, the model may not possess the required proficiency to create highly precise technical content, particularly for specialized fields where areaspecific advanced domain knowledge is required [121]. As an AI-powered assistant, ChatGPT has limitations when it comes to assessing the credibility of its training data, which can affect the accuracy of the information it generates [16]. Moreover, ChatGPT does not have access to the Internet and currently has limited knowledge of world events after 2021 [122,123]. Therefore, some of its responses may be outdated or inaccurate since knowledge is continuously evolving, and its training data set is outdated. For example, when ChatGPT is required to produce up-to-date references for any specific area, ChatGPT may provide fabricated references that seem plausible but are not backed by real-world sources [122].

One drawback of utilizing ChatGPT in the scientific and engineering disciplines for research paper writing will be its potential to lack the creativity and tone that human writers possess. Writing research papers requires more than just providing accurate and consistent data; it also involves effectively conveying complex information in a clear and simplistic manner. The use of ChatGPT may result in less engaging and effective output due to its possible lack of creative and nuanced writing [122]. Another disadvantage is that ChatGPT-based scientific and engineering written content could be biased as per the training dataset and algorithm design, etc., which is one of the biggest problems in AI-powered chatbots because they lack the basic human touch in writing and hence avoid the biases. As per the principle of 'garbage-in-garbage-out,' AI-powered chatbots have the potential to reinforce biases by utilizing large sets of data that already contain these biases [124].

### 5.2 Advanced Plagiarism Detection Based on Counter AI-Algorithms–(Application Layer Enhancement)

In academia and research, the use of ChatGPT as an AI-powered chatbot with the ability to generate text that resembles human language has sparked several ethical concerns related to academic dishonesty [113,119,125]. Although ChatGPT's responses are not exact copies of any specific text, they are generated by synthesizing training data, which may lead to producing responses similar to those of existing sources. A recent test [116] has shown that ChatGPT wrote a 500-word essay that had a 45% similarity to existing sources. The danger is that students might use ChatGPT without realizing the potential for plagiarism [121]. Moreover, ChatGPT's ability to generate research studies and scientific abstracts with fabricated data poses a significant risk to academic integrity [121]. This capability may encourage higher education students to rely solely on ChatGPT when writing academic essays, which leads to academic dishonesty. Furthermore, ChatGPT's tendency to generate incorrect and nonsensical answers raises the risk of misinformation in scientific publications, making this ethical issue even more serious.

As generative-AI based content generation tools are beginning to emerge, proponents will advocate their use as long it is responsibly acknowledged by the student. The AI-chatbot, ChatGPT, is extremely popular, but academics and researchers all over the world are worried that excessive use of ChatGPT in technical research papers and thesis writing [122] will result in sub-optimal research efforts. Academicians and researchers predict that the uncontrolled use of ChatGPT can be

quite catastrophic for the research community in general and the scientific and engineering research fraternity in particular [123]. Some alarming facts show that ChatGPT has demonstrated its capacity to provide satisfactory responses to exam questions in medical and legal fields [118,123]. An empirical study revealed that ChatGPT's answers to life support exams at a university were, on average, relevant and accurate and more closely aligned with guidelines than previous studies that utilized other AI tools. Given this level of performance, ChatGPT poses a significant threat to academic integrity, particularly in higher education [122].

To overcome the problem of plagiarism, counter AI-detection tools like GPTZero [15] are already in use. These tools can predict if a certain text was potentially generated by generative-AI software by analyzing the linguistic constructs in the text [125]. Moreover, quite recently, Turnitin similarity index checking software has also introduced a new option in their software which checks the AI involvement in the written text [11].

#### 5.3 Biometrics-Based Student Authentication and Behavior Prediction-(Session Layer Enhancement)

Authenticating the identity and behavioral patterns of the student studying or taking exams in distant (online mode) is one of the biggest future areas of research for LMS development. Advanced image/video processing techniques backed by AI will be required for biometrics-based student authentication and behavior detection [126–130]. Previously available techniques, such as Face recognition, have often been shown to be easily deceived if a photo of an individual is presented to a camera. The new research direction will focus on diverse student biometrics-based user identification such as fingerprint, handprint, and iris detection using commodity webcam or smartphone cameras. Biometrics-based student authentication will be a big step toward the prevention of academic misconduct in online lecture delivery and exams.

# 5.4 Use of Improved Transport Layer Protocols for Video Conferencing Applications in LMS–(Transport Layer Enhancement)

User Datagram Protocol (UDP) and Transmission Control Protocol (TCP) are the most commonly used transport layer protocols for elastic data delivery on the Internet. This covers data delivery with different QoS requirements, such as latency-sensitive or non-real-time data delivery. Of these, UDP is preferred for real-time data applications, and TCP is used for stored video streaming. TCP ensures reliable data transmission through variable data-rate on the basis of receiver feedback; it does so by retransmitting any lost data packets, making it unsuitable for any real-time video transmission. UDP uses fixed-rate data delivery irrespective of network conditions. Transmitting lower-resolution video quality or a video with many lost frames may compromise the effectiveness of AI-algorithms used for online proctoring and other tasks. AI-algorithms used for advanced video analytics need to work with higher-quality videos. Modern LMS video conferencing applications will use a combination of TCP and UDP for video transmission. TCP will be used for the control plane for probing the network, and UDP will be used in the data plane for flexible rate data delivery, ensuring acceptable QoS. Newer transport protocols designed for video content delivery, such as those used for online lectures through LMS, are becoming increasingly popular. These protocols are equipped with advanced network probing algorithms, such as PathRake TCP [131], which allow for efficient estimation of available bandwidth, even in the presence of firewalls on the network path. The use of these protocols will greatly improve the user's quality of experience.

#### 5.5 Seamless Communication of Smart Cameras with Software Defined Networking (SDN) Framework for Live LMS Sessions–(Network Layer Enhancement)

Legacy routing service on the Internet over TCP/IP protocol is only providing the best-effort service for data delivery on the Internet. Currently, LMS applications use these conventional routing services for video conferencing applications for online classes/exam conduction. The result may be sub-optimal as end users may be connected with variable connection speeds. In addition, network conditions may fluctuate with time. Specialized video conferencing applications, such as Zoom, use peer-to-peer data delivery methods to optimize video transmission, which provides additional network probing service to gauge the network conditions to quickly reroute packets in events of congestion on default routes. LMS applications of the future will instead rely on the Software-Defined Networking (SDN) framework, which gives the network the ability to re-route data packets quickly in the event of Network Failure or if the path does not meet the user QoS requirements, providing seamless service offering dedicated QoS without any involvement of LMS application for network probing service. In the future, segregated data services in the LMS applications requiring different QoS may use different routing services in the network. Real-time video delivery in future LMS will integrate smart cameras instead of regular webcams, which could interact seamlessly with SDN controllers [132] directly while the remaining latency-insensitive non-video data service could use conventional routing services. The SDN controllers will work on adaptive routing protocols to optimize QoS [133] for efficient video delivery to be processed by AI-based image/video processing algorithms at the server end.

#### 6 Discussion

The future of education will likely involve distance-learning (online) techniques facilitated through advanced Learning Management Systems. These changes will be primarily driven by already available advancements in real-time data streaming and enhanced data security and QoS. However, AI-driven advanced applications will be running on top of them to enhance the user learning experience through AI-based adaptive learning strategies tailored to individual needs. These will further be used to credibly authenticate and examine the individuals. AI-backed deep data analytics will help in the evolution of course contents and devise new AI-based adaptive learning strategies to meet course learning objectives and outcomes. However, AI offers equal opportunities for examiners and examinees to benefit from them. Learning Management Systems will always have to be one step ahead of state-of-art techniques and be equipped with counter AI-algorithms.

#### 7 Conclusion

AI-based adaptive learning techniques have long been in development in various civilian, industry, and military sectors due to their proven benefits over conventional training. In particular, the university and vocational education are areas where remarkable benefits have been seen when using AI-based LMS in comparison to conventional modes of lecture. However, the recent COVID-19 pandemic has spurned a renewed interest in addressing previously unanticipated and unexplored research and development challenges. This is because all previous development efforts were made assuming that the trainee could be brought for training into a specialized facility having proprietary hardware and software. However, during COVID-19, the online training had to be imparted to individuals who were geographically separated and connected through heterogeneous technologies involving COTS hardware and open-sourced 3rd party software and various network speeds, causing all sorts of equipment compatibility issues. This article presented an exhaustive literature review covering all facets of AI-based techniques used recently by researchers to address the different challenges mentioned

above. This paper also presented the challenges that are yet to be overcome in the current stateof-the-art research for the development of AI-based LMS. These include challenges of real-time online proctoring solutions for academic misconduct detection, privacy challenges, rate-adaptive video-conferencing applications, the delegation of server-centric and client-centric tasks in the face of internet/electricity outages, plagiarism detection for generative-AI responses, and more well-informed grievance procedures and policies if students challenged any of the AI-driven decisions among others.

To conclude about the strengths and weaknesses of ChatGPT, utilizing it for scientific and engineering research paper writing can save valuable time and energy while enhancing precision and uniformity. Nonetheless, these approaches may produce flawed or erroneous content and fall short in conveying the originality and details which can be covered by human authors. It is essential to carefully consider the advantages and disadvantages of ChatGPT-based methods before implementing them for the generation of technical content.

Acknowledgement: The authors acknowledge the facilities at their respective institutes to facilitate this research.

Funding Statement: The authors received no specific funding for this study.

Author Contributions: The authors confirm their contribution to the paper as follows: study conception and design: Sameer Qazi, Muhammad Bilal Kadri, Muhammad Naveed, Bilal A. Khawaja; data collection: Sameer Qazi, Bilal A. Khawaja; analysis and interpretation of results: Sameer Qazi, Muhammad Bilal Kadri, Muhammad Naveed, Bilal A. Khawaja, Sohaib Zia Khan, Muhammad Mansoor Alam, Mazliham Mohd Su'ud; draft manuscript preparation: Sameer Qazi, Muhammad Bilal Kadri, Bilal A. Khawaja. All authors reviewed the results and approved the final version of the manuscript.

Availability of Data and Materials: This article does not involve data availability and this section is not applicable.

**Conflicts of Interest:** The authors declare that they have no conflicts of interest to report regarding the present study.

#### References

- W. Hussain, W. G. Spady, M. T. Naqash, S. Z. Khan, B. A. Khawaja and L. Conner, "ABET accreditation during and after COVID19—Navigating the digital age," *IEEE Access*, vol. 8, pp. 218997–219046, 2020. doi: 10.1109/ACCESS.2020.3041736.
- [2] S. Jain, "How to proctor exams on an LMS," Apr. 2022, Accessed: Jun. 17, 2023. [Online]. Available: https://blog.mettl.com/lms-proctoring/
- [3] W. Hussain, W. G. Spady, S. Z. Khan, B. A. Khawaja, T. Naqash and L. Conner, "Impact evaluations of engineering programs using ABET student outcomes," *IEEE Access*, vol. 9, pp. 46166–46190, 2021. doi: 10.1109/ACCESS.2021.3066921.
- [4] S. S. Chua, J. B. Bondad, Z. R. Lumapas, and J. D. L. Garcia, "Online examination system with cheating prevention using question bank randomization and tab locking," in 2019 4th Int. Conf. Inf. Technol. (InCIT), Bangkok, Thailand, IEEE, Oct. 2019, pp. 126–131. doi: 10.1109/INCIT.2019.8912065.
- [5] P. Chakraborty, P. Mittal, M. S. Gupta, S. Yadav, and A. Arora, "Opinion of students on online education during the COVID-19 pandemic," *Hum. Behav. Emerg. Tech.*, vol. 3, no. 3, pp. 357–365, Dec. 2020. doi: 10.1002/hbe2.240.

- [6] Microsoft. "Microsoft teams," Accessed: Jun. 22, 2023. [Online]. Available: https://www.microsoft.com/ en-ww/microsoft-teams/group-chat-software
- [7] "Zoom," Accessed: Jun. 18, 2023. [Online]. Available: https://zoom.us/
- [8] "Pearson VUE," Accessed: Mar. 12, 2023. [Online]. Available: https://home.pearsonvue.com/
- [9] "Autoproctor," Accessed: Jul. 20, 2023. [Online]. Available: https://www.autoproctor.co/
- [10] "ProctorU," Accessed: May 13, 2023. [Online]. Available: https://www.proctoru.com/
- [11] "Turnitin," Accessed: Sep. 02, 2023. [Online]. Available: https://www.turnitin.com/
- [12] "IEEE CrossCheck," Accessed: Sep. 02, 2023. [Online]. Available: https://crosscheck.ieee.org/crosscheck/
- [13] "iThenticate," Accessed: Sep. 02, 2023. [Online]. Available: https://www.ithenticate.com/
- [14] J. P. McIntire, L. K. McIntire, and P. R. Havig, "Methods for chatbot detection in distributed text-based communications," in 2010 Int. Symp. Collab. Technol. Syst., Chicago, IL, USA, IEEE, 2010, pp. 463–472. doi: 10.1109/CTS.2010.5478478.
- [15] "GPT zero," Accessed: May 16, 2023. [Online]. Available: https://gptzero.me/
- [16] "OPEN AI GP2 output detector," Accessed: Sep. 02, 2023. [Online]. Available: https://openai-openai-detector.hf.space/
- [17] M. Ahmed, H. U. Khan, and E. U. Munir, "Conversational AI: An explication of few-shot learning problem in transformers-based chatbot systems," *IEEE Trans. Comput. Soc. Syst.*, vol. 11, no. 2, pp. 1888– 1906, Apr. 2024. doi: 10.1109/TCSS.2023.3281492.
- [18] "Contentatscale AI content detector," Accessed: Apr. 12, 2023. [Online]. Available: https://contentatscale. ai/ai-content-detector/
- [19] "Writers AI content detector," Accessed: Aug. 19, 2023. [Online]. Available: https://writer.com/ai-contentdetector/
- [20] D. F. Murad, R. Hassan, Y. Heryadi, B. D. Wijanarko, and Titan, "The impact of the COVID-19 pandemic in Indonesia (face to face versus online learning)," in 2020 Third Int. Conf. Vocat. Educ. Electr. Eng. (ICVEE), Surabaya, Indonesia, IEEE, Oct. 2020, pp. 1–4. doi: 10.1109/ICVEE50212.2020.9243202.
- [21] D. Mondego and E. Gide, "The importance of mentorship to enhance online learning and teaching in higher education in the COVID-19 era: A comprehensive literature review," in 2021 19th Int. Conf. Inf. Technol. Based High. Educ. Train. (ITHET), Sydney, Australia, IEEE, Nov. 2021, pp. 1–4. doi: 10.1109/ITHET50392.2021.9759747.
- [22] T. Supriyatno and F. Kurniawan, "A new pedagogy and online learning system on pandemic COVID 19 era at islamic higher education," in 2020 6th Int. Conf. Educ. Technol. (ICET), Malang, Indonesia, IEEE, Oct. 2020, pp. 7–10. doi: 10.1109/ICET51153.2020.9276604.
- [23] D. L. Edy, Widiyanti, and Basuki, "Revisiting the impact of project-based learning on online learning in vocational education analysis of learning in pandemic COVID-19," in 2020 4th Int. Conf. Vocat. Educ. Train. (ICOVET), Malang, Indonesia, IEEE, Sep. 2020, pp. 378–381. doi: 10.1109/ICOVET50258.2020.9230137.
- [24] S. Z. Salas-Pilco, "The impact of COVID-19 on Latin American STEM higher education: A systematic review," in 2022 IEEE World Eng. Educ. Conf. (EDUNINE), Santos, Brazil, IEEE, Mar. 2022, pp. 1–6. doi: 10.1109/EDUNINE53672.2022.9782354.
- [25] H. Wang, J. Wu, M. Muhedaner, and S. Maihemuti, "College students online education evaluation through swot analysis during COVID-19," *IEEE Access*, vol. 10, pp. 88371–88385, 2022. doi: 10.1109/AC-CESS.2022.3198082.
- [26] H. Hassani, E. S. Silva, S. Unger, M. TajMazinani, and S. Mac Feely, "Artificial intelligence (AI) or intelligence augmentation (IA): What is the future?" AI, vol. 1, no. 2, pp. 143–155, Apr. 2020. doi: 10.3390/ai1020008.
- [27] V. Sze, Y. H. Chen, T. J. Yang, and J. S. Emer, "Efficient processing of deep neural networks: A tutorial and survey," *Proc. IEEE*, vol. 105, no. 12, pp. 2295–2329, Dec. 2017. doi: 10.1109/JPROC.2017.2761740.
- [28] Z. Zhou, X. Chen, E. Li, L. Zeng, K. Luo and J. Zhang, "Edge intelligence paving the last mile of artificial intelligence with edge computing," *Proc. IEEE*, vol. 107, no. 8, pp. 1738–1762, Aug. 2019. doi: 10.1109/JPROC.2019.2918951.

- [29] E. Tjoa and C. Guan, "A survey on explainable artificial intelligence (XAI): Toward medical XAI," *IEEE Trans. Neural Netw. Learn. Syst.*, vol. 32, no. 11, pp. 4793–4813, Nov. 2021. doi: 10.1109/TNNLS.2020.3027314.
- [30] S. Qazi, B. A. Khawaja, and Q. U. Farooq, "IoT-equipped and AI-enabled next generation smart agriculture: A critical review, current challenges and future trends," *IEEE Access*, vol. 10, pp. 21219– 21235, 2022. doi: 10.1109/ACCESS.2022.3152544.
- [31] K. Shafique, B. A. Khawaja, F. Sabir, S. Qazi, and M. Mustaqim, "Internet of things (IoT) for nextgeneration smart systems: A review of current challenges, future trends and prospects for emerging 5G-IoT scenarios," *IEEE Access*, vol. 8, pp. 23022–23040, 2020. doi: 10.1109/ACCESS.2020.2970118.
- [32] H. K. M. AL-Chalabi, A. M. A. Hussein, and U. C. Apoki, "An Adaptive learning system based on learner's knowledge level," in 2021 13th Int. Conf. Electron., Comput. Artif. Intell. (ECAI), Pitesti, Romania, IEEE, Jul. 2021, pp. 1–4. doi: 10.1109/ECAI52376.2021.9515158.
- [33] N. Nenkov, G. Dimitrov, Y. Dyachenko, and K. Koeva, "Artificial intelligence technologies for personnel learning management systems," in 2016 IEEE 8th Int. Conf. Intell. Syst. (IS), Sofia, Bulgaria, IEEE, Sep. 2016, pp. 189–195. doi: 10.1109/IS.2016.7737420.
- [34] J. Skrebeca, P. Kalniete, J. Goldbergs, L. Pitkevica, D. Tihomirova and A. Romanovs, "Modern development trends of chatbots using artificial intelligence (AI)," in 2021 62nd Int. Sci. Conf. Inf. Technol. Manag. Sci. Riga Tech. University (ITMS), Riga, Latvia, IEEE, Oct. 2021, pp. 1–6. doi: 10.1109/ITMS52826.2021.9615258.
- [35] N. Albayrak, A. Ozdemir, and E. Zeydan, "An overview of artificial intelligence based chatbots and an example chatbot application," in 2018 26th Signal Process. Commun. Appl. Conf. (SIU), Izmir, IEEE, May 2018, pp. 1–4. doi: 10.1109/SIU.2018.8404430.
- [36] A. Deveci Topal, C. Dilek Eren, and A. Kolburan Geçer, "Chatbot application in a 5th grade science course," *Educ. Inf. Technol.*, vol. 26, no. 5, pp. 6241–6265, Sep. 2021. doi: 10.1007/s10639-021-10627-8.
- [37] Y. Y. Yang and B. Shulruf, "An expert-led and artificial intelligence system-assisted tutoring course to improve the confidence of Chinese medical interns in suturing and ligature skills: A prospective pilot study," J. Educ. Eval. Health Prof., vol. 16, pp. 7, Apr. 2019. doi: 10.3352/jeehp.2019.16.7.
- [38] D. G. Krishnan, "Artificial intelligence in oral and maxillofacial surgery education," Oral Maxillofac. Surg. Clin. North Am., vol. 34, no. 4, pp. 585–591, Nov. 2022. doi: 10.1016/j.coms.2022.03.006.
- [39] F. Schwendicke *et al.*, "Artificial intelligence for oral and dental healthcare: Core education curriculum," *J. Dent.*, vol. 128, no. 4, pp. 104363, Jan. 2023. doi: 10.1016/j.jdent.2022.104363.
- [40] A. M. Rahman, A. A. Mamun, and A. Islam, "Programming challenges of chatbot: Current and future prospective," in 2017 IEEE Reg. 10 Humanit. Technol. Conf. (R10-HTC), Dhaka, Bangladesh, IEEE, Dec. 2017, pp. 75–78. doi: 10.1109/R10-HTC.2017.8288910.
- [41] "OpenAI ChatGPT," Accessed: Aug. 15, 2023. [Online]. Available: https://openai.com/blog/chatgpt
- [42] J. Zhang et al., "HiVeGPT: Human-machine-augmented intelligent vehicles with generative pretrained transformer," *IEEE Trans. Intell. Veh.*, vol. 8, no. 3, pp. 2027–2033, Mar. 2023. doi: 10.1109/TIV.2023.3256982.
- [43] Y. Bengio, Y. Lecun, and G. Hinton, "Deep learning for AI," Commun. ACM, vol. 64, no. 7, pp. 58–65, Jul. 2021. doi: 10.1145/3448250.
- [44] S. M. Kelly, "ChatGPT passes exams from law and business schools," CNN Business. Jan. 26, 2023, Accessed: Jun. 25, 2023. [Online]. Available: https://edition.cnn.com/2023/01/26/tech/chatgpt-passesexams/index.html
- [45] M. R. King and chatGPT, "A conversation on artificial intelligence, chatbots, and plagiarism in higher education," *Cel. Mol. Bioeng.*, vol. 16, no. 1, pp. 1–2, Feb. 2023. doi: 10.1007/s12195-022-00754-8.
- [46] B. R. Ranoliya, N. Raghuwanshi, and S. Singh, "Chatbot for university related FAQs," in 2017 Int. Conf. Adv. Comput., Commun. Inform. (ICACCI), Udupi, India, IEEE, Sep. 2017, pp. 1525–1530. doi: 10.1109/ICACCI.2017.8126057.

- [47] Y. Ugurlu and M. Karabulut, "Artificial intelligence supported turkish university virtual assistant," in 2021 Innov. Intell. Syst. Appl. Conf. (ASYU), Elazig, Turkey, IEEE, Oct. 2021, pp. 1–5. doi: 10.1109/ASYU52992.2021.9599003.
- [48] N. Wang, "Application of artificial intelligence and virtual reality technology in the construction of university physical education," in 2022 3rd Int. Conf. Electron. Commun. Artif. Intell. (IWECAI), Zhuhai, China, IEEE, Jan. 2022, pp. 343–346. doi: 10.1109/IWECAI55315.2022.00073.
- [49] Y. H. V. Chiang, Y. R. Lin, and N. S. Chen, "Using deep learning models to predict student performance in introductory computer programming courses," in 2022 Int. Conf. Adv. Learn. Technol. (ICALT), Bucharest, Romania, IEEE, Jul. 2022, pp. 180–182. doi: 10.1109/ICALT55010.2022.00060.
- [50] A. S. Aljaloud *et al.*, "A deep learning model to predict student learning outcomes in LMS using CNN and LSTM," *IEEE Access*, vol. 10, pp. 85255–85265, 2022. doi: 10.1109/ACCESS.2022.3196784.
- [51] Ö. F. Akmeşe, H. Kör, and H. Erbay, "Use of machine learning techniques for the forecast of student achievement in higher education," *Inf. Technol. Learn. To.*, vol. 82, no. 2, pp. 297–311, Apr. 2021. doi: 10.33407/itlt.v82i2.4178.
- [52] L. Slusky, "Cybersecurity of online proctoring systems," J. Int. Technol. Inf. Manag., vol. 29, no. 1, pp. 56–83, Jan. 2020. doi: 10.58729/1941-6679.1445.
- [53] "Eklavya," Accessed: Mar. 12, 2023. [Online]. Available: https://www.eklavya.com/Online-Examination. html
- [54] "Respondus," Accessed: May 15, 2023. [Online]. Available: https://web.respondus.com/
- [55] "Talview: AI powered hiring and proctoring solution," Accessed: May 12, 2023. [Online]. Available: https://www.talview.com/
- [56] A. Nigam, R. Pasricha, T. Singh, and P. Churi, "A systematic review on AI-based proctoring systems: Past, present and future," *Educ. Inf. Technol.*, vol. 26, no. 5, pp. 6421–6445, Sep. 2021. doi: 10.1007/s10639-021-10597-x.
- [57] T. K. F. Chiu, Q. Xia, X. Zhou, C. S. Chai, and M. Cheng, "Systematic literature review on opportunities, challenges, and future research recommendations of artificial intelligence in education," *Comput. Educ.: Artif. Intell.*, vol. 4, no. 3, pp. 100118, 2023. doi: 10.1016/j.caeai.2022.100118.
- [58] S. Z. Salas-Pilco, "The impact of AI and robotics on physical, social-emotional and intellectual learning outcomes: An integrated analytical framework," *Br. J. Educ. Technol.*, vol. 51, no. 5, pp. 1808–1825, Sep. 2020. doi: 10.1111/bjet.12984.
- [59] A. Vahabzadeh, N. Keshav, R. Abdus-Sabur, K. Huey, R. Liu and N. Sahin, "Improved socio-emotional and behavioral functioning in students with autism following school-based smartglasses intervention: Multi-stage feasibility and controlled efficacy study," *Behav. Sci.*, vol. 8, no. 10, pp. 85, Sep. 2018. doi: 10.3390/bs8100085.
- [60] P. J. Standen *et al.*, "An evaluation of an adaptive learning system based on multimodal affect recognition for learners with intellectual disabilities," *Br J. Educ. Technol.*, vol. 51, no. 5, pp. 1748–1765, Sep. 2020. doi: 10.1111/bjet.13010.
- [61] T. McCarthy, L. P. Rosenblum, B. G. Johnson, J. Dittel, and D. M. Kearns, "An artificial intelligence tutor: A supplementary tool for teaching and practicing braille," *J. Vis. Impair. Blin.*, vol. 110, no. 5, pp. 309–322, Sep. 2016. doi: 10.1177/0145482X1611000503.
- [62] H. Sun, X. Wang, and X. Wang, "Application of blockchain technology in online education," Int. J. Emerg. Technol. Learn., vol. 13, no. 10, pp. 252–259, Oct. 2018. doi: 10.3991/ijet.v13i10.9455.
- [63] G. Chen, B. Xu, M. Lu, and N. S. Chen, "Exploring blockchain technology and its potential applications for education," *Smart Learn. Environ.*, vol. 5, no. 1, pp. 1, Dec. 2018. doi: 10.1186/s40561-017-0050-x.
- [64] W. Villegas-Ch., S. Sánchez-Viteri, and M. Román-Cañizares, "Academic activities recommendation system for sustainable education in the age of COVID-19," *Informatics*, vol. 8, no. 2, pp. 29, Apr. 2021. doi: 10.3390/informatics8020029.
- [65] K. Sharma, Z. Papamitsiou, and M. Giannakos, "Building pipelines for educational data using AI and multimodal analytics: A 'grey-box' approach," *Br. J. Educ. Technol.*, vol. 50, no. 6, pp. 3004–3031, Nov. 2019. doi: 10.1111/bjet.12854.

- [66] S. C. Tsai, C. H. Chen, Y. T. Shiao, J. S. Ciou, and T. N. Wu, "Precision education with statistical learning and deep learning: A case study in Taiwan," *Int. J. Educ. Technol. High Educ.*, vol. 17, no. 1, pp. 12, Dec. 2020. doi: 10.1186/s41239-020-00186-2.
- [67] "Moodle," Accessed: Feb. 24, 2023. [Online]. Available: https://moodle.org/
- [68] "Canvas," Accessed: Apr. 12, 2023. [Online]. Available: https://www.instructure.com/canvas
- [69] "Blackboard," Accessed: Apr. 12, 2023. [Online]. Available: https://www.blackboard.com/en-eu/teachinglearning/learning-management
- [70] "ProctorFree," Accessed: May 13, 2023. [Online]. Available: https://www.proctorfree.com/lms
- [71] "HonorLock," Accessed: Jun. 17, 2023. [Online]. Available: https://honorlock.com/blog/lms-integrationwith-online-proctoring-software/
- [72] "Mercer Mettl," Accessed: Jul. 15, 2023. [Online]. Available: https://mettl.com/en/
- [73] R. N. Tarigan, R. Nadlifatin, and A. P. Subriadi, "Academic dishonesty (cheating) in online examination: A literature review," in 2021 Int. Conf. Comput. Sci., Inf. Technol., Electr. Eng. (ICOMITEE), Banyuwangi, Indonesia, IEEE, Oct. 2021, pp. 148–153. doi: 10.1109/ICOMITEE53461.2021.9650082.
- [74] S. G. Rabiha, I. H. Kartowisastro, R. Setiawan, and W. Budiharto, "Survey of online exam proctoring model to detect cheating behavior based on face recognition," in 2022 8th Int. Conf. Syst. Inform. (ICSAI), Kunming, China, IEEE, Dec. 2022, pp. 1–7. doi: 10.1109/ICSAI57119.2022.10005488.
- [75] S. Kavitha, S. Mohanavalli, and B. Bharathi, "Predicting learning behaviour of online course learners' using hybrid deep learning model," in 2018 IEEE 6th Int. Conf. MOOCs, Innov. Technol. Educ. (MITE), Hyderabad, India, IEEE, Nov. 2018, pp. 98–102. doi: 10.1109/MITE.2018.8747136.
- [76] B. Q. Cui and C. R. Zhou, "Design of online teaching quality evaluation system for private university:— Research based on deep learning algorithm," in 2020 Int. Conf. Modern Educ. Inf. Manag. (ICMEIM), Dalian, China, IEEE, Sep. 2020, pp. 29–32. doi: 10.1109/ICMEIM51375.2020.00015.
- [77] Z. Li, Z. Zhu, and T. Yang, "A multi-index examination cheating detection method based on neural network," in 2019 IEEE 31st Int. Conf. Tools Artif. Intell. (ICTAI), Portland, OR, USA, IEEE, Nov. 2019, pp. 575–581. doi: 10.1109/ICTAI.2019.00086.
- [78] M. A. Samir, Y. Maged, and A. Atia, "Exam cheating detection system with multiple-human pose estimation," in 2021 IEEE Int. Conf. Comput. (ICOCO), Kuala Lumpur, Malaysia, IEEE, Nov. 2021, pp. 236–240. doi: 10.1109/ICOCO53166.2021.9673534.
- [79] S. Essahraui, M. A. El Mrabet, M. F. Bouami, K. E. Makkaoui, and A. Faize, "An intelligent anti-cheating model in education exams," in 2022 5th Int. Conf. Adv. Commun. Technol. Netw. (CommNet), Marrakech, Morocco, IEEE, Dec. 2022, pp. 1–6. doi: 10.1109/CommNet56067.2022.9993953.
- [80] I. N. Yulita, E. Paulus, A. Sholahuddin, and D. Novita, "AdaBoost support vector machine method for human activity recognition," in 2021 Int. Conf. Artif. Intell. Big Data Anal., Bandung, Indonesia, IEEE, Oct. 2021, pp. 1–4. doi: 10.1109/ICAIBDA53487.2021.9689713.
- [81] B. Li and E. Li, "Automated online proctoring system using gaze view tracking and custom object detection," in 2022 IEEE MIT Undergrad. Res. Technol. Conf. (URTC), Cambridge, MA, USA, IEEE, Sep. 2022, pp. 1–5. doi: 10.1109/URTC56832.2022.10002245.
- [82] N. Dilini, A. Senaratne, T. Yasarathna, N. Warnajith, and L. Seneviratne, "Cheating detection in browserbased online exams through eye gaze tracking," in 2021 6th Int. Conf. Inf. Technol. Res. (ICITR), Moratuwa, Sri Lanka, IEEE, Dec. 2021, pp. 1–8. doi: 10.1109/ICITR54349.2021.9657277.
- [83] A. Gupta and A. Bhat, "Bluetooth camera based online examination system with deep learning," in 2022 6th Int. Conf. Intell. Comput. Control Syst. (ICICCS), Madurai, India, IEEE, May 2022, pp. 1477–1480. doi: 10.1109/ICICCS53718.2022.9788147.
- [84] S. Rudian, M. H. Khan, and N. Pinkwart, "Using H5P in exams: A method to prevent cheating," in 2021 Int. Conf. Adv. Learn. Technol. (ICALT), Tartu, Estonia, IEEE, Jul. 2021, pp. 36–37. doi: 10.1109/ICALT52272.2021.00018.
- [85] G. Kasliwal, "Cheating detection in online examinations," Master of Science, San Jose State University, San Jose, CA, USA, 2015. doi:10.31979/etd.y292-cddh.

- [86] R. Bawarith, A. Basuhail, A. Fattouh, and S. Gamalel-Din, "E-exam cheating detection system," Int. J. Adv. Comput. Sc., vol. 8, no. 4, 2017. doi: 10.14569/IJACSA.2017.080425.
- [87] Y. Atoum, L. Chen, A. X. Liu, S. D. H. Hsu, and X. Liu, "Automated online exam proctoring," *IEEE Trans. Multimed.*, vol. 19, no. 7, pp. 1609–1624, Jul. 2017. doi: 10.1109/TMM.2017.2656064.
- [88] A. Kumar, M. Madaan, S. Kumar, A. Saha, and K. Gupta, "Handwriting generation and synthesis using recurrent neural networks," in 2022 13th Int. Conf. Comput. Commun. Netw. Technol. (ICCCNT), Kharagpur, India, IEEE, Oct. 2022, pp. 1–8. doi: 10.1109/ICCCNT54827.2022.9984450.
- [89] F. KreB *et al.*, "Hardware-aware workload distribution for AI-based online handwriting recognition in a sensor pen," in 2022 11th Mediterr. Conf. Embed. Comput. (MECO), Budva, Montenegro, IEEE, Jun. 2022, pp. 1–4. doi: 10.1109/MECO55406.2022.9797131.
- [90] B. Rajesh, A. K. Gupta, A. Raj, and M. Javed, "HWRCNet: Handwritten word recognition in JPEG compressed domain using CNN-BiLSTM network," 2022. doi: 10.48550/ARXIV.2201.00947.
- [91] F. D. Pereira *et al.*, "Towards human-AI collaboration: A recommender system to support CS1 instructors to select problems for assignments and exams," *IEEE Trans. Learn. Technol.*, vol. 16, no. 3, pp. 1–14, 2022. doi: 10.1109/TLT.2022.3224121.
- [92] "BeyondTrust privilege management," Accessed: Mar. 15, 2023. [Online]. Available: https://www. beyondtrust.com/
- [93] M. Liwicki, M. Weber, T. Zimmermann, and A. Dengel, "Seamless integration of handwriting recognition into pen-enabled displays for fast user interaction," in 2012 10th IAPR Int. Workshop Doc. Anal. Syst., Gold Coast, Queenslands, TBD, Australia, IEEE, Mar. 2012, pp. 364–368. doi: 10.1109/DAS.2012.78.
- [94] "SonicWall: Next-gen firewalls & cybersecurity solutions," Accessed: May 17, 2023. [Online]. Available: https://www.sonicwall.com/
- [95] N. Selwyn, C. O'Neill, G. Smith, M. Andrejevic, and X. Gu, "A necessary evil? The rise of online exam proctoring in Australian universities," *Media Int. Austr.*, vol. 186, no. 1, pp. 149–164, Feb. 2023. doi: 10.1177/1329878X211005862.
- [96] J. V. Henry and M. Oliver, "Who will watch the watchmen? The ethico-political arrangements of algorithmic proctoring for academic integrity," *Postdigit. Sci. Educ.*, vol. 4, no. 2, pp. 330–353, Apr. 2022. doi: 10.1007/s42438-021-00273-1.
- [97] V. Echeverria, K. Yang, L. Lawrence, N. Rummel, and V. Aleven, "Designing hybrid human-AI orchestration tools for individual and collaborative activities: A technology probe study," *IEEE Trans. Learn. Technol.*, vol. 16, no. 2, pp. 191–205, Apr. 1 2023. doi: 10.1109/TLT.2023.3248155.
- [98] J. Zhou and C. -M. Pun, "Personal privacy protection via irrelevant faces tracking and pixelation in video live streaming," *IEEE Trans. Inform. Forensic Secur.*, vol. 16, pp. 1088–1103, 2021. doi: 10.1109/TIFS.2020.3029913.
- [99] M. Labayen, R. Vea, J. Flórez, N. Aginako, and B. Sierra, "Online student authentication and proctoring system based on multimodal biometrics technology," *IEEE Access*, vol. 9, pp. 72398–72411, 2021. doi: 10.1109/ACCESS.2021.3079375.
- [100] S. Coghlan, T. Miller, and J. Paterson, "Good proctor or "Big Brother"? AI ethics and online exam supervision technologies," 2020. doi: 10.48550/ARXIV.2011.07647.
- [101] A. M. Elshamy, M. A. Abdelghany, A. Q. Alhamad, H. F. A. Hamed, H. M. Kelash and A. I. Hussein, "Secure implementation for video streams based on fully and permutation encryption techniques," in 2017 Int. Conf. Comput. Appl. (ICCA), Doha, United Arab Emirates, IEEE, Sep. 2017, pp. 50–55. doi: 10.1109/COMAPP.2017.8079738.
- [102] B. Boyadjis, C. Bergeron, B. Pesquet-Popescu, and F. Dufaux, "Extended selective encryption of H.264/AVC (CABAC)- and HEVC-encoded video streams," *IEEE Trans. Circuits Syst. Video Technol.*, vol. 27, no. 4, pp. 892–906, Apr. 2017. doi: 10.1109/TCSVT.2015.2511879.
- [103] G. van Wallendael, A. Boho, J. de Cock, A. Munteanu, and R. van de Walle, "Encryption for high efficiency video coding with video adaptation capabilities," *IEEE Trans. Consum. Electron.*, vol. 59, no. 3, pp. 634–642, Aug. 2013. doi: 10.1109/TCE.2013.6626250.

- [104] T. Sikandar, W. N. A. W. Samsudin, M. F. Rabbi, and K. H. Ghazali, "An efficient method for detecting covered face scenarios in ATM surveillance camera," *SN Comput. Sci.*, vol. 1, no. 3, pp. 151, May 2020. doi: 10.1007/s42979-020-00163-6.
- [105] A. M. Leach, N. Ammar, D. N. England, L. M. Remigio, B. Kleinberg and B. J. Verschuere, "Less is more? Detecting lies in veiled witnesses," *Law Hum. Behav.*, vol. 40, no. 4, pp. 401–410, Aug. 2016. doi: 10.1037/lhb0000189.
- [106] A. A. S. Alashbi and M. S. Sunar, "Occluded face detection, face in Niqab dataset," in *Emerging Trends in Intelligent Computing and Informatics, Advances in Intelligent Systems and Computing*, F. Saeed, F. Mohammed, N. Gazem (Eds.), Cham: Springer International Publishing, 2020, vol. 1073, pp. 209–215, doi:10.1007/978-3-030-33582-3\_20.
- [107] A. Sharma, "Top proctoring software for online exams," Feb. 24, 2023, Accessed: Jun. 30, 2023, [Online]. Available: https://www.techjockey.com/blog/proctoring-software-for-online-exam
- [108] L. R. Solano-Hurtado and M. M. Soto-Cordova, "A study on video streaming quality of DASH scheme in multimedia services," in 2021 Congreso Internacional de Innovación y Tendencias en Ingeniería (CONIITI), Bogotá, Colombia, IEEE, Sep. 2021, pp. 1–4. doi: 10.1109/CONIITI53815.2021.9619720.
- [109] P. Chakraborty, S. Dev, and R. H. Naganur, "Dynamic HTTP live streaming method for live feeds," in 2015 Int. Conf. Comput. Intell. Commun. Netw. (CICN), Jabalpur, India, IEEE, Dec. 2015, pp. 1394–1398. doi: 10.1109/CICN.2015.333.
- [110] G. J. Sullivan, J. R. Ohm, W. J. Han, and T. Wiegand, "Overview of the high efficiency video coding (HEVC) standard," *IEEE Trans. Circuits Syst. Video Technol.*, vol. 22, no. 12, pp. 1649–1668, Dec. 2012. doi: 10.1109/TCSVT.2012.2221191.
- [111] "Mininet," Accessed: Jul. 15, 2023. [Online]. Available: http://mininet.org/
- [112] "OpenVswitch," Accessed: Jul. 14, 2023. [Online]. Available: https://www.openvswitch.org/
- [113] Z. Zou, Y. Jin, P. Nevalainen, Y. Huan, J. Heikkonen and T. Westerlund, "Edge and fog computing enabled AI for IoT-An overview," in 2019 IEEE Int. Conf. Artif. Intell. Circuits Syst. (AICAS), Hsinchu, Taiwan, IEEE, Mar. 2019, pp. 51–56. doi: 10.1109/AICAS.2019.8771621.
- [114] I. Y. Jung and H. Y. Yeom, "Enhanced security for online exams using group cryptography," *IEEE Trans. Educ.*, vol. 52, no. 3, pp. 340–349, Aug. 2009. doi: 10.1109/TE.2008.928909.
- [115] A. N. Varnavsky, "Chatbot to increase the effectiveness of the «Flipped Classroom» technology," in 2022 2nd Int. Conf. Technol. Enhanced Learn. Higher Educ. (TELE), Lipetsk, Russian Federation, IEEE, May 2022, pp. 289–293. doi: 10.1109/TELE55498.2022.9801001.
- [116] T. Brown et al., "Language models are few-shot learners," in Advances in Neural Information Processing Systems, Curran Associates, Inc., 2020, pp. 1877–1901. [Online]. Available: https://proceedings.neurips.cc/ paper\_files/paper/2020/file/1457c0d6bfcb4967418bfb8ac142f64a-Paper.pdf
- [117] E. Kasneci et al., "ChatGPT for good? On opportunities and challenges of large language models for education," *Learn. Individ. Differ.*, vol. 103, no. 1, pp. 102274, Apr. 2023. doi: 10.1016/j.lindif.2023.102274.
- [118] J. Qadir, "Engineering education in the era of ChatGPT: Promise and pitfalls of generative AI for education," in 2023 IEEE Glob. Eng. Educ. Conf. (EDUCON), Kuwait, IEEE, May 2023, pp. 1–9. doi: 10.1109/EDUCON54358.2023.10125121.
- [119] Y. K. Dwivedi *et al.*, "Opinion Paper: "So what if ChatGPT wrote it?" Multidisciplinary perspectives on opportunities, challenges and implications of generative conversational AI for research, practice and policy," *Int. J. Inf. Manag.*, vol. 71, no. 2, pp. 102642, Aug. 2023. doi: 10.1016/j.ijinfomgt.2023.102642.
- [120] D. R. E. Cotton, P. A. Cotton, and J. R. Shipway, "Chatting and cheating: Ensuring academic integrity in the era of ChatGPT," *Innov. Educ. Teach. Int.*, pp. 1–12, Mar. 2023. doi: 10.35542/osf.io/mrz8h.
- [121] J. Rudolph, S. Tan, and S. Tan, "ChatGPT: Bullshit spewer or the end of traditional assessments in higher education?" J. Appl. Learn. Teach., vol. 6, no. 1, Jan. 2023. doi: 10.37074/jalt.2023.6.1.9.
- [122] M. Farrokhnia, S. K. Banihashem, O. Noroozi, and A. Wals, "A SWOT analysis of ChatGPT: Implications for educational practice and research," *Innov. Educ. Teach. Int.*, pp. 1–15, Mar. 2023. doi: 10.1080/14703297.2023.2195846.

- [123] A. Lecler, L. Duron, and P. Soyer, "Revolutionizing radiology with GPT-based models: Current applications, future possibilities and limitations of ChatGPT," *Diagn. Interv. Imaging*, vol. 104, no. 6, pp. 269–274, Feb. 2023. doi: 10.1016/j.diii.2023.02.003.
- [124] P. P. Ray, "ChatGPT: A comprehensive review on background, applications, key challenges, bias, ethics, limitations and future scope," *Internet Things Cyber-Phys. Syst.*, vol. 3, no. 1, pp. 121–154, 2023. doi: 10.1016/j.iotcps.2023.04.003.
- [125] D. Mhlanga, "Open AI in education, the responsible and ethical use of ChatGPT towards lifelong learning," in *FinTech and Artificial Intelligence for Sustainable Development*, 2023, pp. 387–409. doi: 10.1007/978-3-031-37776-1\_17.
- [126] C. K. Lo, "What is the impact of ChatGPT on education? A rapid review of the literature," *Educ. Sci.*, vol. 13, no. 4, pp. 410, Apr. 2023. doi: 10.3390/educsci13040410.
- [127] N. Fijačko, L. Gosak, G. Štiglic, C. T. Picard, and M. John Douma, "Can ChatGPT pass the life support exams without entering the American heart association course?" *Resuscitation*, vol. 185, no. 1, pp. 109732, Apr. 2023. doi: 10.1016/j.resuscitation.2023.109732.
- [128] J. Liang, "SEL fusion system: Multisource digital biometrics and stimuli for early ASD screening," in 2023 IEEE 6th Int. Conf. Inf. Syst. Comput. Aided Educ. (ICISCAE), Dalian, China, IEEE, Sep. 2023, pp. 776–782. doi: 10.1109/ICISCAE59047.2023.10393755.
- [129] N. Ruiz et al., "ATL-BP: A student engagement dataset and model for affect transfer learning for behavior prediction," *IEEE Trans. Biom. Behav. Identity Sci.*, vol. 5, no. 3, pp. 411–424, Jul. 2023. doi: 10.1109/TBIOM.2022.3210479.
- [130] A. Kumar and Y. Zhou, "Human identification using finger images," *IEEE Trans. Image Process.*, vol. 21, no. 4, pp. 2228–2244, Apr. 2012. doi: 10.1109/TIP.2011.2171697.
- [131] N. Ito, T. Oshiba, K. Satoda, and T. Murase, "PathRakeTCP: Available bandwidth estimation using multiple TCP connections for passing through firewalls," in 2023 IEEE Int. Conf. Consum. Electron. (ICCE), Las Vegas, NV, USA, IEEE, Jan. 2023, pp. 1–6. doi: 10.1109/ICCE56470.2023.10043399.
- [132] O. Kochan, M. Beshley, H. Beshley, Y. Shkoropad, I. Ivanochko and N. Seliuchenko, "SDN-based internet of video things platform enabling real-time edge/cloud video analytics," in 2023 17th Int. Conf. Exp. Design. Appl. CAD Syst. (CADSM), Jaroslaw, Poland, IEEE, Feb. 2023, pp. 1–5. doi: 10.1109/CADSM58174.2023.10076539.
- [133] T. S. Andjamba and G. A. L. Zodi, "A load balancing protocol for improved video on demand in SDNbased clouds," in 2023 17th Int. Conf. Ubiquitous Inf. Manag. Commun. (IMCOM), Seoul, Republic of Korea, IEEE, Jan. 2023, pp. 1–6. doi: 10.1109/IMCOM56909.2023.10035591.